**Author’s response to reviews**

**Title:** How do stakeholders from multiple hierarchical levels of a large provincial health system define engagement? A qualitative study

**Authors:**
Jill Norris (jmnorris@ucalgary.ca)
Deborah White (dwhit@ucalgary.ca)
Lorelli Nowell (lnowell@ucalgary.ca)
Kelly Mrklas (kelly.mrklas@albertahealthservices.ca)
Henry Stelfox (tstelfox@ucalgary.ca)

**Version:** 2  **Date:** 23 May 2017

**Author’s response to reviews:**

Reviewer reports:

Thank you for revising your manuscript, however, the Editor would like to request for another round of revision. Kindly see the comments below for your reference:

"please provide only a tracked change version outlining additions and exclude elements that are simply moving the text to a different place. This will help provide a clean copy for us to base a decision on"

* We have excluded tracking for text that was moved. All tracked changes in the submission are additions.